Author's response to reviews

Title: Intramedullary neurocysticercosis presenting as Brown Sequard Syndrome

Authors:

Elda M Salazar (massiels24@gmail.com)
Fernando J Escoto (escotomd@gmail.com)
Rita Pineda (ritapineda6@gmail.com)

Version: 3 Date: 28 August 2014

Author's response to reviews:

Dear Editor:

The changes made to the previous manuscript have been the following:
- Line numbers were included.
- Abstract was structured in three sections.
- Clarification about the patient written informed consent of the publication of this report and clinical images was made.
- Acknowledgement, Abbreviation and authors' contributions section was included.
- two more pictures were included.

Thank you,

Elda. M Salazar
Hospital General San Juan de Dios
Guatemala, Guatemala
massiels24@gmail.com
phone: 50257526564